Author's response to reviews

Title: Caveolin-2 associates with intracellular chlamydial inclusions independently of caveolin-1

Authors:

Wilmore C Webley (wilmore@microbio.umass.edu)
Leonard C Norkin (lnorkin@microbio.umass.edu)
Elizabeth S Stuart (esstuart@microbio.umass.edu)

Version: 4 Date: 19 July 2004

Author's response to reviews: see over
Dear Dr. Borthwick,

We have reviewed our manuscript MS: 1411745315335340, entitled “Caveolin-2 associates with intracellular chlamydial inclusions independently of caveolin-1” in compliance with the BMC Infectious Disease format.

Specifically, we have removed all the unnecessary capitalizations from the title of the manuscript as well as from the submission system. On the title page we have included the country of the affiliated institution where the work was conducted. We have also formatted the abstract dividing it in the required sections of Background, Methods, Results and Conclusion. The competing interests section has also been completed.

The references were formatted for BMC Infectious Disease using EndNote 7 and the manuscript will therefore be uploaded in Microsoft Word (doc. Format; XP professional). The figures were formatted into separate PDF files and uploaded as well. The Figure legends were placed immediately after the references and the table and table-legend was placed after the figure legends.

Please let us know if there is anything else that would be needed to make this manuscript ready for publication.

Yours Sincerely,

Wilmore Webley, Ph.D.